Author’s response to reviews

Title: Small-cell Neuroendocrine Carcinoma of the Ileum: Case Report

Authors:

Jong Eun Lee (elduke00@schmc.ac.kr)
Sung Hoon Hong (soondaeman00@naver.com)
Hae Il Jung (gs2834@schmc.ac.kr)
Myoung Won Son (mwson@schmc.ac.kr)
Tae sung Ahn (eyetoeeye@schmc.ac.kr)
Sun Wook Han (cheia@schmc.ac.kr)
JunHun Cho (jununiu@schmc.ac.kr)

Version: 2 Date: 07 Aug 2019

Author’s response to reviews:

1. Initial CT scan attach to Figure part.
2. Overlapped section has been changed.
3. Author's contributions have been changed.
4. Clarify contents for publication was also obtained for accompanying images.
5. Two section to be in lines.
6. Not applicable
7. The consent and CARE form were be removed.
8. Manuscript was uploaded as a clean version.
9. Some changes were made as your mentioned.

Thank you so much.